Author’s response to reviews

Title: A novel missense mutation in the MYH7 gene causes an uncharacteristic phenotype of myosin storage myopathy: a case report

Authors:

Jean Mamelona (Jean.Mamelona@vitalitenb.ca)
Louisa Filice (Louisa.Filice@usherbrooke.ca)
Yousef Oussedik (Yousef.Oussedik@vitalitenb.ca)
Nicolas Crapoulet (Nicolas.Crapoulet@vitalitenb.ca)
Alier Marrero (Alier.Marrero@vitalitenb.ca)

Version: 5 Date: 05 Apr 2019

Author’s response to reviews:

1. Thank you for providing the updated accession number for your data. However, when we used this number to search for your data, it still did not yield any results. Please ensure that the accession number is correct and that the data is released before publication. This is because it will not be possible to make any changes to the availability of data and materials section once your article has been accepted for publication.

The accession number is correct. The data is now public

(Please check this link: https://www.ncbi.nlm.nih.gov/clinvar/variation/560881/)

2. Please relocate the additional files list to after the references.

Done properly. The additional files list has been moved to after the references.

Thank you!